Author's response to reviews

Title: Reliability, construct and discriminative validity of clinical testing in subjects with and without chronic neck pain

Authors:

René Jørgensen RJ (rejo@ucsyd.dk)
Inge Hansen IR (iris@health.sdu.dk)
Deborah Falla DF (deborah.falla@bccn.uni-goettingen.uni.de)
Birgit J Kristensen BJK (bjuul-kristensen@health.sdu.dk)

Version: 2 Date: 3 June 2014

Author's response to reviews:

Thank you very much for your comments to the manuscript

Minor changes have been made on you request - following bellow.

1. On the title page, email addresses of all authors are now displayed together with the nationality of the authors.

2. In the methods section, under the subheading sample size, a sentence saying that the study conformed to the Helsinki declaration have been added to paragraph covering ethical approval details. Information on informed consent was already present in the same section.

I hope these changes are sufficient.

Kind regards

René